**Author’s response to reviews**

**Title:** Combining serological and contact data to derive target immunity levels for achieving and maintaining measles elimination

**Authors:**

Sebastian Funk (Sebastian.Funk@LSHTM.ac.uk)

Jennnifer Knapp (ylf9@cdc.gov)

Emmaculate Lebo (ipc8@cdc.gov)

Susan Reef (ser2@cdc.gov)

Alya Dabbagh (dabbagha@who.int)

Katrina Kretsinger (kretsingerk@who.int)

Mark Jit (mark.jit@lshtm.ac.uk)

W. John Edmunds (john.edmunds@lshtm.ac.uk)

Peter Strebel (pstrebel@gavi.org)

**Version:** 2 **Date:** 16 Aug 2019

**Author’s response to reviews:**

Dear Editor,

Thank you for the assessment of our revised manuscript. In response to the reviewer, we have now added a supplementary figure showing the mean estimated contact-adjusted immunity as a function of the number of bootstrap samples used. We refer to this in the paper where we state the number of bootstrap samples used.

Regards,

Sebastian Funk